Author's response to reviews


Authors:

John Bond (john.bond@newcastle.ac.uk)
Janet Wilson (janet.wilson@nuth.nhs.uk)
Martin Eccles (martin.eccles@newcastle.ac.uk)
Alessandra Vanoli (alessandra_vanoli@yahoo.it)
Nick Steen (nick.steen@newcastle.ac.uk)
Ray Clarke (rayclarke@aol.com)
Andrew Zarod (apzarod@aol.com)
Catherine Lock (c.a.lock@newcastle.ac.uk)
Katie Brittain (katie.brittain@newcastle.ac.uk)
Chris Speed (chris.speed@newcastle.ac.uk)
Nikki Rousseau (n.s.rousseau@newcastle.ac.uk)

Version: 3 Date: 9 August 2006

Author's response to reviews:

Dear Editorial Production Team

Once again thank you for agreeing to publish our manuscript.

The following changes have been made:

The Abstract has been structured into Background, Methods/Design, Discussion and has been updated on the submission page.

Discussions have been added to both the Abstract and the main body of text.

I hope these changes meet with your approval.

Yours sincerely

Catherine Lock